Author's response to reviews

Title: Dermatofibrosarcoma of the breast: a case report

Authors:

Olivier F Cottier (olivier_cottier@hotmail.com)
Maryse Fiche (Maryse.Fiche@chuv.ch)
Jean-Yves Meuwly (Jean-Yves.Meuwly@chuv.ch)
Jean-François Delaloye (Jean-Francois.Delaloye@chuv.ch)

Version: 3 Date: 9 April 2011

Author's response to reviews: see over
Dear Editors

I would like to submit an original work about a Dermatofibrosarcoma protuberans of the breast. It is a rare neoplasm of soft tissue and its location in the breast is extremely uncommon. The aim of this case report is to highlight the importance of the differential diagnosis of such kind of lesion that can mimic a primary breast tumor. New surgical procedures in case of postoperative discovery of Dermatofibrosarcoma protuberans is recommended to ensure local control of disease.

According to your recommendations for authors, we submit the article online.

Best Regards

Docteur Olivier Cottier
Chef de clinique,
Service gynécologie-obstétrique